Author's response to reviews

Title: A case study of outsourced primary healthcare services in Sindh, Pakistan: Is this a real reform?

Authors:

Sana Tanzil (syeda.sana@aku.edu)
Aysha Zahidie (aysha.zahidie@aku.edu)
Adeel Ahsan (adeelahsan83@gmail.com)
Ambreen Kazi (akamran@KSU.EDU.SA)
Babar T Shaikh (shaikh.babar@gmail.com)

Version: 4 Date: 13 June 2014

Author's response to reviews: see over
Covering letter

Query:
Ethics - Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/e/policy/b3.htm), and any experimental research on animals must follow internationally recognized guidelines. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Response:
Details added and highlighted with yellow.

Query:
Consent - Informed consent must also be documented. Please revise you manuscript to clearly state whether you obtained written informed consent from the study participants.

Response:
Details added and highlighted with yellow.